Author's response to reviews

Title: Risk-reducing strategies for women carrying BRCA1/2 mutations with a focus on prophylactic surgery

Authors:

Mohamed Salhab (msalhab1@gmail.com)
Selina Bismohun (m0600131@sgul.ac.uk)
kefah Mokbel (kefahmokbel@hotmail.com)

Version: 6 Date: 19 October 2010

Author's response to reviews:

Dear editor

Thank you for accepting our article for publication.

Please find attached the final version of the manuscript with the formatting changes requested by BMC editorial production team.

Kind regards

Mohamed Salhab
On behalf of the co authors